Author's response to reviews

Title: An unusual presentation of anetoderma: a case report

Authors:

Shahin Aghaei (shahinaghaei@yahoo.com)
Manoochehr Sodaify (sodaifym@sums.ac.ir)
Fatemeh Sari Aslani (Sariasf@sums.ac.ir)
Nazila Mazharinia (namany03@yahoo.com)

Version: 5 Date: 27 July 2004

Author's response to reviews: see over
Dear Dr Parkin

Thank you and the editorial team of BMC Dermatology for accepting the manuscript.

All of the changes that have been made in the revised final version of the manuscript are as following:

1-The title was corrected to: An unusual presentation of anetoderma: a case report

2-The page numbers were removed.

3-The sections were changed as BMC order.

4-Funding sources: None

Truly yours

Shahin Aghaei